Author’s response to reviews

Title: Diagnostic accuracy of touch imprint cytology for head and neck malignancies: A useful intra-operative tool in resource limited countries.

Authors:

Hania Naveed (hana_fahim@yahoo.com)
Mariam Abid (mariamabid77@yahoo.com)
Atif Hashmi (doc_atif2005@yahoo.com)
Muhammad Muzzammil Edhi (muhammad_edhi@brown.edu)
Ahmareen Sheikh (drahmareen@hotmail.com)
Ghazala Mudasir (ghazalamudassir@yahoo.com)
Amir Khan (dramirkhan04@gmail.com)

Version: 1 Date: 30 Aug 2017

Author’s response to reviews:

Editor's comments:

The Introduction and Discussion sections of the manuscript are too long and should be made more concise.

Ans: Introduction and discussion sections are made concise now as per your suggestion.

Please confirm the period of the study: from July 2015 to March 2016 it is not 1 year and 3 months, but 9 months.

Ans. Study duration is from 1st July 2015 to 30th march 2016 for a period of 9 months. Also edited in the manuscript.

The Table in the Material and Methods section should not be included in the body of the text.

Ans. Corrected.
Thanks.

Regards,

Corresponding Author.